Author's response to reviews

Title: Significance of MDR1 and multiple drug resistance in refractory human epileptic brain

Authors:

Nicola Marchi (marchi@marionegri.it)
Kerri L Hallene (hallenk@ccf.org)
Kelly M Kight (kightk@ccf.org)
Luca Cucullo (cuculll@ccf.org)
Gabriel Moeddel (moeddeg@ccf.org)
Gabriele Dini (dinig@ccf.org)
William Bingaman (bingamb@ccf.org)
Annamaria Vezzani (vezzani@marionegri.it)
Damir Janigro (janigrd@ccf.org)

Version: 6 Date: 27 September 2004

Author's response to reviews:

Ladies and Gentlemen:

Please find in attachment the revised version of the manuscript entitled "Significance of MDR1 and multiple drug resistance in refractory human epileptic brain". All the changes included in the copyedited version have been reviewed and accepted, and only a minor formatting change has been made (font in the acknowledgement section, page 20 of the revised version).

Best regards,

Damir Janigro, Ph.D.